Author's response to reviews

Title: Murraya koenigii leaf extract inhibits proteasome activity and induces cell death in breast cancer cells

Authors:

Bindu Noolu (bindu.noolu@gmail.com)
Rajanna Ajumeera (arajanna@gmail.com)
Anitha Chauhan (anithachauhan@yahoo.com)
Balakrishna Nagalla (dr_nbk@yahoo.com)
Raghunath Manchala (mraghunath55@yahoo.com)
Ayesha Ismail (ayeshai2005@gmail.com)

Version: 3 Date: 20 December 2012

Author's response to reviews: see over
To,
Dr. Tom Rowles,
Executive Editor,
BMC Complementary & Alternative Medicine

Subject: MS: 1602273541770919 - Murraya koenigii leaf extract inhibits proteasome activity and induces cell death in breast cancer cells

Dear Dr. Rowles,

I would like to thank you for in principle acceptance of our manuscript for publication in the journal BMC Complementary & Alternative Medicine.

The manuscript was sent with editorial comments only and these have been incorporated in the revised version. Please find below point-by-point responses to the editorial comments.

**Editorial Comments:**

**Comment:** Please could you confirm who identified the curry leaves used in your study and include this information in your manuscript.

**Answer:** The identity of the curry leaves was confirmed by Dr. B. Pratibha Devi, Professor & Head Department of Botany, Osmania University, Hyderabad, India. This information is added in the methods section of the manuscript on pg 7.

**Comment:** Similarly, could you please confirm whether a voucher specimen of your plant has been deposited in a herbarium, and include this information in your manuscript.

**Answer:** A voucher specimen has been deposited in the herbarium of the above mentioned department at the Osmania University, Hyderabad, India. This information is added in the methods section of the manuscript on pg 7.

Please note that the manuscript (text, tables and figures) has been formatted according to the journal format. I hope you find the revised manuscript in order and suitable for publication in BMC Complementary and Alternative Medicine at the earliest.

Best regards,

(Ayesha Ismail)
Corresponding Author